Author's response to reviews

Title: Hardship financing of healthcare among rural poor in Orissa, India

Authors:

Erika Binnendijk (erikabin@nendijk@gmail.com)
Ruth Koren (r.koren@post.tau.ac.il)
David M. Dror (daviddror@socialre.org)

Version: 4 Date: 20 January 2012

Author's response to reviews:

In response to your email dated 19 January 2012, we have removed all references to NGOs and funding bodies approving the study for its ethical standards. We have instead included a statement that there was no local ethics committee in place in Orissa, India at the time you conducted the study, however we held a two day workshop during which the ethical aspects of the study were discussed.